Author’s response to reviews

Title: Analysis of eight genes modulating interferon gamma and human genetic susceptibility to tuberculosis: a case-control association study

Authors:

Marlo Moller (marlom@sun.ac.za)
Almut Nebel (a.nebel@mucosa.de)
Paul D van Helden (pvh@sun.ac.za)
Stefan Schreiber (s.schreiber@mucosa.de)
Eileen G Hoal (egvh@sun.ac.za)

Version: 4 Date: 2 June 2010

Author’s response to reviews:

The abstract is now the same in both "manuscript details" and in the manuscript.

No other changes were made to the manuscript